Familial Hypercholesterolemia: Therapeutic Advances

Ashok K Singh

Department of Pharmaceutical Sciences, Babasaheb Bhimrao Ambedkar University, Vidya Vihar, Raebareli Road, Lucknow 226025, India

Abstract: Familial hypercholesterolemia (FH), the most common and severe form of hypercholesterolemia, is an autosomal codominant disease characterized by an increased plasma low density lipoprotein (LDL)-cholesterol and premature atherosclerosis. The most common cause of FH is the mutations either in LDL receptor, apolipoprotein B (ApoB), or proprotein convertase subtilisin kexin type 9 (PCSK9) genes. However, it is now known that many subjects with severe inherited hypercholesterolemia have no defects in these genes. These cases are caused either by mutations in genes yet to be identified or are consequences of polygenic, epigenetic, or acquired defects. Management of FH is based on risk factor modification and use of multiple lipid-lowering medications. Lipoprotein apheresis concomitant with lipid-lowering therapy is the treatment of choice for homozygous FH, preferably with LDL-C levels > 200 mg/dl (>300 mg/dl if without CAD). A microsomal triglyceride transfer protein inhibitor and an antisense oligonucleotide against ApoB have recently been approved for use in subjects with clinically diagnosed homozygous familial hypercholesterolemia. Anti-PCSK9 monoclonal antibodies showed encouraging results and are currently being evaluated in phase III clinical trials. Early identification of affected individuals and aggressive treatment should significantly reduce the burden of cardiovascular disease in society. The aim of this short review is to describe the new frontier of PCSK9 inhibition.

Keywords: Familial Hypercholesterolemia, Atherosclerosis, Coronary heart diseases (CHD), Recent therapeutic strategies.

1. Introduction

The low density lipoprotein receptor (LDLR) binds low density lipoprotein (LDL) at the cell surface and internalizes LDL by receptor- mediated endocytosis. Mutations in the LDLR gene, leading to defective LDLRs and disrupted clearance of LDL, cause familial hypercholesterolemia. Typically, familial hypercholesterolemia heterozygotes have plasma LDL cholesterol levels in the range of 6-11 mmol/l, whereas homozygotes have plasma LDL cholesterol levels of approximately 20 mmol/l [1]. Importantly defects in at least three different genes encoding for proteins involved in hepatic clearance of LDL-c from the plasma can cause FH. Mutations in the apolipoproteina B-100 (APOB) and in the proprotein convertase subtilisin/kexin 9 (PCSK9) genes can produce a phenotype of FH. Homozygous FH (homo-FH) cases have two mutant alleles and heterozygous-FH (hetero-FH) cases have one mutant and one normal allele. FH is caused by mutations of one of three FH associated genes (FH genes): LDL-receptor (LDLR), proprotein convertase subtilisin/kexin type 9 (PCSK9), or apolipoprotein B-100 (ApoB). Hetero- and homo-FH can be clinically differentiated by the severity of several features such as generalized xanthomatosis, hypercholesterolemia and CHD. Homo-FH patients are likely to be identified as young children because of the early occurrence of xanthomatosis associated with exceptionally high plasma cholesterol concentrations exceeding 13.00 mmol/L [2].

The pathogenesis of FH was evaluated by Brown & Goldstein, who discovered mutations in the LDLR gene in FH patients. Recently many gene mutations (including 1741 LDLR gene mutations) that cause FH have been deposited in public databases. After the discovery of LDLR mutations in FH patients, other less common causes of FH have been found, such as mutations in ApoB, which encodes the apoB protein that is a LDLR ligand and gain-of-function (g-o-f) mutations of PCSK9. Homo-FH patients with a mild phenotype often show the phenotype of plasma cholesterol

concentrations similar to hetero-FH. Thus, the usefulness of molecular genetic diagnosis in differentiating homo-FH and hetero-FH should be evaluated, and could potentially be used to reduce the number of underdiagnosed FH cases [3]. Although the diagnostic algorithm of FH currently involves the screening of the LDLR, APOB and PCSK9 genes, molecular studies have been showing that in a high number of clinically defined index-cases no pathogenic mutations can be identified at these three loci. On the other hand, the role of some genetic variants described in the previously referred loci (LDLR, APOB and PCSK9), remains to be established [4].

Moreover, Familial Combined Hyperlipidemia (FCH) is the most common genetic dyslipidemia with estimated prevalence approximately 1% in the general population. It is associated with a number of metabolic disorders such as hypertension, obesity, insulin resistance, diabetes and metabolic syndrome and it is an important cause of premature CAD [5].

Furthermore, in case of FH, CHD is influenced by different cardiovascular risk factors such as, age, gender, weight, blood pressure, LDL-C and high density lipoprotein cholesterol (HDL-C) plasma values, mutations in LDL-R, and diet. Moreover, it is thought that oxidative stress (OS) plays a particularly important role in the development of cardiovascular diseases [6]. (Raquel et al)

Recent Therapeutic Approaches

Statins reduce the risk for cardiovascular events and progression of atherosclerosis in FH. 57 Adjuvant therapy with ezetimibe, bile acid sequestrants, plant stanols or sterols, and niacin may often also be required. In case of hypertriglyceridemia, use of fenofibrate or omega-3 fish oils may be advisable, [7] and niacin could also be considered when plasma LDL-cholesterol are also not at target [8]. Probucol is a potent antioxidant and, despite lowering HDL

cholesterol, can regress xanthomata and reduce CVD events in FH. Statins and ezetimibe can lower LDL cholesterol in patients with homozygous FH by potentially decreasing hepatic secretion of apoB, but these patients usually require apheresis [9]. If apheresis is not available, consideration should be given to the addition of lomitapide or mipomersen therapy to further lower LDL cholesterol. It is critical to ensure that all patients with FH follow the recommended treatment regimens. Patients should be counseled on all aspects of their care, including the warning signs of drugrelated toxicity. [10] With the hope of providing additional approaches to lowering LDL-C levels, many agents which differ in actions are being investigated.

Microsomal transfer protein (MTP) inhibitors

Microsomal Triglyceride Transfer Protein (MTP), found in the endoplasmic reticulum of hepatocytes and enterocytes, has been identified as one of the promising targets for the treatment of hepercholesterolemia. MTP plays crucial role in the assembly of triglyceride rich chylomicrons in enterocytes, and VLDL in hepatocytes. Inhibition of MTP thereby leads to decrease of VLDL-C, LDL-C and TG levels. This is why MTP inhibitors could be helpful in treating FH. Lomitapide [AEGR-733, Juxtapid, Aegerion Pharmaceuticals, Cambridge, MA) is an oral microsomal triglyceride transfer protein inhibitor that decreases the hepatic production and secretion of VLDL. Lomitapide is licensed for the treatment of homozygous FH in the United States and Europe as an add-on therapy. The use of lomitapide may also result in significant reductions in other lipids and lipoproteins, including total cholesterol, apoB, triglyceride, and non-HDL cholesterol. Lomitapide may be hepatotoxic, can elevate plasma aminotransferases, and can increase intrahepatic fat content by approximately 6% after 26 and 78 weeks of therapy. It was also concluded that cytochrome P450 3A4 inhibitors increase the exposure to lomitapide [11]. Lomitapide may reduce the absorption of fat-soluble vitamins and essential fatty acids, SO coadministration of appropriate supplements is recommended. Because of the risk of hepatotoxicity, in the United States, lomitapide must also be prescribed through an FDA approved Risk Evaluation Mitigation Strategy program. However, mipomersen and lomitapide are both FDA approved for the treatment of patients with homozygous FH. (Gerald F. Watts et al) Several other MTP inhibitors are in various stages of development. Such candidates are CP-346086 [12], JTT-130 [13], SLx-4090 [14], implitapide [15], dirlotapide [16] and its analogues.

Antisense oligonucleotides to Apo B

ApoB is required for the intracellular assembly and secretion of very low density lipoproteins (VLDL) and LDL by the liver. Thus, the number of apoB is positively correlated with the level of plasma LDL-C. Therefore, a possible afford has been taken to inhibit the formation of apoB during the process of translation from the gene into its protein product [17]. A potential approach to inhibit translation of mRNA is to block the process by using a single-strand antisense oligonucleotide (ASO) that is complementary to and will strongly hybridize to the mRNA. This will lead to the degradation of mRNA and eventually result in reduced transcription of the encoded protein. ApoB ASO targeting apoB100 protein synthesis is an attractive ASO since apoB100 is highly distributed in the liver [18]. Current research mainly focuses on apoB100 ASO, with mipomersen monotherapy. [Mipomersen is an apolipoprotein B synthesis inhibitor for lowering of LDL-C in patients who are already receiving lipid-lowering drugs, including high-dose statins.] [19].

Mipomersen

Mipomersen (Kynamro; Genzyme, Cambridge, MA) is an antisense 20-mer oligonucleotide that binds to a complementary sequence messenger RNA encoding apoB, thereby inhibiting ribosomal translation. By inhibiting the biosynthesis of apoB, hepatic very-low-density lipoprotein (VLDL) production and secretion are significantly reduced. Mipomersen consists of a phosphorotioate backbone and 20-O-(2-methoxyethyl)-modified ends, which provide biological stability. Subsequent to subcutaneous injection, mipomersen is concentrated in the liver, where it undergoes catabolism via the action of hepatic endonucleases and exonucleases [20]. Mipomersen is FDA approved for use in patients with homozygous FH. Mipomersen has been shown to reduce serum LDL cholesterol by approximately 25%, 28%, and 36% in patients with homozygous FH, heterozygous FH, and severe hypercholesterolemia with or without CHD, respectively. Mipomersen also induces substantial reductions in total cholesterol, apoB, triglycerides, non-HDL cholesterol, and Lp(a). In addition to frequent injection site reactions and short-lived fatigue and myalgia, mipomersen can induce hepatic steatosis in 16% of patients, as well as elevations in plasma aminotransferases in 8% of patients. These hepatic changes apparently resolve on discontinuing the drug [21, 22]. Mipomersen has orphan drug status and, because of the risk of hepatotoxicity, can only be prescribed in the United States through a Risk Evaluation Mitigation Strategy program [21, 22, 23].

Proprotein Convertase Subtilisin Kexin type 9 inhibition

Proprotein convertase subtilisin/kexin type 9, also known as PCSK9, is an enzyme that in humans is encoded by the PCSK9 gene. This gene encodes a proprotein convertase belonging to the proteinase K subfamily of the secretary subtilase family. The encoded protein plays a major role in cholesterol homeostasis. Thus, drugs that block PCSK9 can lower circulating cholesterol. Although, the exact mechanism by which PCSK9 affects LDLR is yet to be determined, PCSK9 acts by targeting the LDLR to lysosomes in a process that involves a direct protein-protein interaction with the receptor and does not require its catalytic activity [24, 25]. Interestingly, statins and ezetimibe treatment were associated with increased levels of circulating PCSK9, which possibly attenuate the therapeutic effects of them. Thus, PCSK9 is considered by many to be a highly desirable therapeutic target for the generation of novel cholesterol-lowering drugs, or in combination with statins and ezetimibe to enhance the lipid-lowering efficacy [26].

PCSK9 inhibitors are not yet licensed for use in FH, but clinical trial data suggest that they may have broad

International Journal of Science and Research (IJSR) ISSN (Online): 2319-7064 Index Copernicus Value (2013): 6.14 | Impact Factor (2013): 4.438

applications for patients with heterozygous FH who are not at LDLcholesterol targets on maximal statin therapy [27, 28] or who are intolerant to statins [29]. At least six different human mAbs and three gene-silencing approaches are under development. Among the mAb developed against PCSK9, clinical trial results are available for the three of them, alirocumab (SAR236553/REGN727), evolocumab (AMG145) and bococizumab (RN316/PF-04950615), while for RG7652, 1B20 and LGT209 no data were published yet. A large body of evidence is available so far for alirocumab and evolocumab. In adition, the association of alirocumab with fibrates slightly affected the efficacy of the antibody in decreasing LDL-C levels. An important consideration relates to the ability of anti-PSCK9 mAbs to decrease the Lp(a) plasma levels. Future studies are needed to characterize the mechanism underlying this effect. PCSK9 inhibitors might therefore represent an optimal therapeutic option to improve the lipid profile of subjects at high CVD risk characterized by elevated Lp(a) plasma levels. Phase I data with the siRNA ALN-PCS were recently released [30], and showed that the inhibition of PCSK9 synthesis by RNA interference (RNAi) provides a potentially safe mechanism to reduce LDL cholesterol concentration in healthy individuals with raised cholesterol. These results set the stage for further of assessment ALN-PCS in patients with hypercholesterolemia, including those treated with statins [31].

Thyromimetics

Thyroid hormone enhances the expression of the hepatic LDLR gene [32], thereby, increasing LDL clearance and decreasing plasma LDL-C levels [33]. Thyroid hormone (in particular 3,5,3'- triiodo-l-thyronine (T3) has, therefore, been tested for use as a cholesterol-lowering agent, but was associated with adverse effects on heart and bones [34]. It is known that T3 exerts its effects via four known isomorphs of the thyroid receptor (TR), TR- α -1, TR- α -2, TR- β -1 and TR- β -2. Further, TR- α plays a major role in the heart, while TR- β , highly expressed in the liver, controls cholesterolemia by mediating the activation of CYP7A in response to T3 [35, 36]. Subsequently, selectively targeting TR- β without the cardiac complications has been developed. Currently, one such agent eprotirome (KB2115), a selective agonist of TR- β , is undergoing clinical trial [37].

Lipoprotein apheresis and other invasive therapies

Lipoprotein apheresis (LA) is an extracorporeal treatment which removes apoB-containing lipoproteins from the circulation [38]. The removal of LDL by LA improves CHD outcomes, progression of atherosclerosis, aortic fibrosis, and endothelial function, in patients with FH [39]. LA is an FDA-approved therapy that is indicated for patients with homozygous or severe heterozygous FH with progressive CHD who are refractory to maximal pharmacotherapy [40]. By contrast to homozygous FH, the patient acceptability and cost-effectiveness of LA for treating refractory heterozygous FH remains unclear and needs further evaluation. Untreated patients with a homozygous phenotype typically have plasma LDL cholesterol 13 mmol/L and should be treated with maximally tolerated pharmacotherapy for at least 6 months before considering LA [39]. Untreated heterozygotes typically have plasma LDL cholesterol from 5 to 13 mmol/L and may be truly nonresponders to or be intolerant of pharmacotherapy [39, 40]. LDL-cholesterol criteria for selecting the aforementioned patients for apheresis have been recommended elsewhere [41] but should be modified according to clinical context. Simple criteria are a reduction in 50 % LDL-cholesterol [39, 41]. Different thresholds for LDL cholesterol may be set according to the availability of resources for apheresis. Imaging should be carried out at baseline to assess aortic stenosis and aneurysms. Contraindications to apheresis methods that use heparin include hemorrhagic diatheses, resistance to adequate coagulation, and hypersensitivity to heparin. LA is efficacious, tolerable, and safe in the treatment of severe FH and may be commenced in children older than of 5 years of age,[42] or earlier in exceptional circumstances. Although lower body weight is a recognized risk factor for complications, successful outcomes have been reported with LA in very young children with homozygous FH [42, 43]. Women with severe FH may be successfully treated during pregnancy. There are several LA methods that are selective for LDL and Lp(a) levels by 50% to 70% after a single treatment [44]. The FDA-approved methods involve the extracorporeal precipitation of apoB-containing lipoprotein with dextran sulfate or heparin, whereas in other countries, alternative systems (immunoadsorption, double cascade filtration or hemoperfusion with direct absorption of lipoproteins, dextran sulfate or polyacrylate) are available [45]. Angiotensin-converting enzyme inhibitors are contraindicated with most systems, particularly the dextran sulfate LDL absorption and hemoperfusion methods because of bradykinin reactions. Patients who are intolerant of a particular method of LA should be tested on an alternative method, including plasma exchange if required. Long-term efficacy of treatment on carotid and aortic valve atherosclerosis should be assessed every 2 years in homozygotes using standard imaging methods [40]. Lomitapide should be considered as an adjunctive treatment to further lower LDL cholesterol in adults with homozygous FH on LA, [46] as well as in children and adolescents with homozygous FH on LA with rapidly progressive atherosclerosis. In those who cannot tolerate lomitapide, mipomersen should be accordingly considered. Novel LDL cholesterol-lowering therapies may reduce the need for or the frequency of LA in severe FH,[47-49] but this needs to be demonstrated. Orthotopic liver transplantation should also be considered in younger homozygous patients when LA is not available, or cannot be tolerated, and LDL cholesterol cannot be adequately controlled with intensive pharmacotherapy [50]. Coronary artery bypass surgery, aortic valve replacement, or a combined heart transplantation should be considered according to clinical context before liver transplantation. Partial ileal bypass should be considered in heterozygous patients who are drugintolerant [51-53]. There may be a future role for gene therapy in treating severe FH [54, 55].

2. Conclusion

It is clear that more than 70% of the body's cholesterol is derived from the de novo cholesterol biosynthesis. Thereby, the inhibition of de novo cholesterol biosynthesis by statins is currently the most effective therapeutic approach to

LDL-C. Statins. However, reduce plasma such recommendations are questionable because muscular side effects and drug interactions of statins are now better understood. Inspite of several adverse effects, statins have blockbuster fame, and the novel hypolipidemic drugs have a task difficult to achieve because their hypolipidemic effect should be better or at least comparable to statins and in addition, toxicity, drug interations and side effects should be minimal. Thus, In addition to existing treatments, some other aforementioned approaches for the development of novel familial hypercholesterolemic treatments are discussed herein. But none of these approaches are currently approved for use in humans. Several ongoing agents are in their different stages of clinical trials, in expectation of promising antihyperlipidemic drugs.

References

- [1] T.B. Strom, K. Tveten, J.K. Laerdahl, T.P. Leren, 2014. Mutation G805R in the transmembrane domain of the LDL receptor gene causes familial hypercholesterolemia by inducing ectodomain cleavage of the LDL receptor in the endoplasmic reticulum. FEBS Open Bio, 4: 321–327.
- [2] R. Fellin, M. Arca, G. Zuliani, S. Calandra, S. Bertolini, 2014. The history of Autosomal Recessive Hypercholesterolemia (ARH). From clinical observations to gene identification. Gene xxx: xxx-xxx.
- [3] H. Mabuchi, A. Nohara, T. Noguchi, J. Kobayashi, M. Kawashiri, I. Takeshi, M. Mori, H.Tada, C. Nakanishi, K. Yagi, M. Yamagishi, K. Ueda, T. Tadayoshi, S. Miyamo, A. Inazu, J. Koizumi, The Hokuriku FH Study Group, 2014. Genotypic and phenotypic features in homozygous familial hypercholesterolemia caused by proprotein convertase subtilisin/ kexin type 9 (PCSK9) gain-of-function mutation. Atherosclerosis 236: 54-61.
- [4] T. Cymbron, P. Mendes, A. Ramos, M. Raposo, N. Kazachkova, A. M. Medeiros, J. Bruges-Armas, M. Bourbon, M. Lima, 2014. Familial hypercholesterolemia: Molecular characterization of possible cases from the Azores Islands (Portugal). Meta Gene 2: 638–645.
- [5] J. Skoumas, C. Liontou, C. Chrysohoou, C. Masoura, K. Aznaouridis, C. Pitsavos, C. Stefanadis, 2014. Statin therapy and risk of diabetes in patients with heterozygous familial hypercholesterolemia or familial combined hyperlipidemia. Atherosclerosis, 237: 140-145.
- [6] R. Cortes, S. Martinez-Hervas, C. Ivorra, G. De Marco, V. Gonzalez-Albert, G. Rojo-Martínez, G. Saez, R. Carmena, J.F. Ascaso, J.T. Real, F.J. Chaves, 2014. Enhanced reduction in oxidative stress and altered glutathione and thioredoxin system response to unsaturated fatty acid load in familial hypercholesterolemia. Clinical Biochemistry, xxx xxxxxx.
- [7] M.J. Chapman, H.N. Ginsberg, P. Amarenco, 2011. Triglyceride-rich lipoproteins and high-density lipoprotein cholesterol in patients at high risk of cardiovascular disease: evidence and guidance for management. Eur Heart J., 32: 1345–1361.
- [8] M.J. Chapman, J.S. Redfern, M.E. McGovern, P. Giral, 2010. Niacin and fibrates in atherogenic dyslipidemia:

Pharmacotherapy to reduce cardiovascular risk. Pharmacol Ther., 126: 314–345.

- [9] T.R. Bates, V.M. Connaughton, G.F. Watts, 2009. Nonadherence to statin therapy: a major challenge for preventive cardiology. Expert Opin Pharmacother., 10: 2973–2985.
- [10] V. Senior, T. Marteau, J. Weinman, 2004. Self-reported adherence to cholesterol-lowering medication in patients with familial hypercholesterolaemia: the role of illness perceptions. Cardiovasc Drugs Ther., 18: 475– 481.
- [11] M. Cuchel, D.J. Rader, 2013. Microsomal transfer protein inhibition in humans. Curr Opin Lipidol., 24: 246.
- [12] D.E.Chandler, Wilder, J.L. Pettini, Y.E. Savoy, S.F. Petras, G. Chang et al, 2003 CP-346086:an MTP inhibitor that lowers plasma cholesterol and triglycerides in experimental animals and in humans. J Lipid Res, 44: 1887-901.
- [13] T. Hata, Y. Mera, T. Kawai, Y. Ishii, Y. Kuroki K. Kakimoto K et al, 2011. Jtt-130, a novel intestinespecific inhibitor of microsomal triglyceride transfer protein, ameliorates impaired glucose and lipid metabolism in zucker diabetic fatty rats. Diabetes. Obes Metab,13: 629–38.
- [14] E. Kim, S. Campbell, O. Schueller, E. Wong, B. Cole, J. Kuo *et al*, 2011. A small-molecule inhibitor of enterocytic microsomal triglyceride transfer protein, slx-4090: biochemical, pharmacodynamic, pharmacokinetic, and safety profile. J Pharmacol Exp Ther, 337: 775–85.
- [15] K. Ueshima, H. Akihisa-Umeno, A. Nagayoshi, S. Takakura, M. Matsuo, S. Mutoh, 2005. Implitapide, a microsomal triglyceride transfer protein inhibitor, reduces progression of atherosclerosis in apolipoprotein E knockout mice fed a Western-type diet: involvement of the inhibition of postprandial triglyceride elevation. Biol Pharm Bull, 28: 247-52.
- [16] J.A. Wren, V.L. King, S.L. Campbell, M.A. Hickman, 2007. Biologic activity of dirlotapide, a novel microsomal triglyceride transfer protein inhibitor, for weight loss in obese dogs. J Vet Pharmacol Ther, 30: 33-42.
- [17] A. Brautbar, C.M. Ballantyne, 2011. Pharmacological strategies for lowering LDL cholesterol: statins and beyond. Nat Rev Cardiol, 8: 253–65.
- [18] R.Z. Yu, T.W. Kim, A. Hong, T.A. Watanabe, H.J. Gaus, R.S. Geary, 2007. Cross-species pharmacokinetic comparison from mouse to man of a second-generation antisense oligonucleotide, isis 301012, targeting human apolipoprotein b-100. Drug Metab Dispos, 35: 460–8.
- [19] F.J. Raal, R.D. Santos, D.J. Blom, A.D. Marais, M.J. Charng, W.C. Cromwell, et al,2010. Mipomersen, an apolipoprotein B synthesis inhibitor, for lowering of LDL cholesterol concentrations in patients with homozygous familial hypercholesterolaemia: a randomised, double-blind, placebo-controlled trial. Lancet, 20: 998-1006.
- [20] M.E. Visser, J.L.Witztum JL, Stroes ESG, Kastelein JJP. Antisense oligonucleotides for the treatment of dyslipidaemia. Eur Heart J. 2012;33:1451–1458.
- [21]E.A. Stein, R. Dufour, C. Gagne, et al, 2012. Apolipoprotein B synthesis inhibition with mipomersen

in heterozygous familial hypercholesterolemia clinical perspective: results of a randomized, double-blind, placebocontrolled trial to assess efficacy and safety as add-on therapy in patients with coronary artery disease. Circulation,126: 2283–2292.

- [22] M.P. McGowan, J-C. Tardif, R. Ceska R, et al, 2012. Randomized, placebocontrolled trial of mipomersen in patients with severe hypercholesterolemia receiving maximally tolerated lipid-lowering therapy. PLoS One. 7: e49006.
- [23] F.J. Raal, R.D. Santos, D.J. Blom, et al, 2010. Mipomersen, an apolipoprotein B synthesis inhibitor, for lowering of LDL cholesterol concentrations in patients with homozygous familial hypercholesterolaemia: a randomised, double-blind, placebo-controlled trial. Lancet, 375: 998–1006.
- [24] D.A. Marais, D.J. Blom, F. Petrides, Y. Gouëffic, G. Lambert, 2012. Proprotein convertase subtilisin/kexin type 9 inhibition. Curr Opin Lipidol, 23: 511-7.
- [25] D.W. Zhang, T.A. Lagace, R. Garuti, Z. Zhao, M. McDonald, J.D. Horton, et al, 2007. Binding of proprotein convertase subtilisin/kexin type 9 to epidermal growth factor-like repeat a of low density lipoprotein receptor decreases receptor recycling and increases degradation. J Biol Chem, 282: 18602–12.
- [26] Y.G. Ni, S. Marco, J.H. Condra, L.B. Peterson, W. Wang, F. Wang, et al, 2010. A proprotein convertase subtilisin-like/kexin type 9 (pcsk9)-binding antibody that structurally mimics the egf (a) domain of LDLreceptor reduces free circulating pcsk9 and LDLcholesterol. J Lipid Res.
- [27] G.K. Hovingh, M.H. Davidson, J.J.P. Kastelein, A.M. O'Connor, 2013. Diagnosis and treatment of familial hypercholesterolaemia. Eur Heart J, 34: 962–971.
- [28] A.S. Wierzbicki, A. Viljoen, T.C. Hardman, D.P. Mikhailidis, 2013. New therapies to reduce low-density lipoprotein cholesterol. Curr Opin Cardiol, 28: 452–457.
- [29] D. Sullivan, A.G. Olsson, R. Scott, et al, 2012. Effect of a monoclonal antibody to pcsk9 on low-density lipoprotein cholesterol levels in statin-intolerant patients: the GAUSS Randomized TrialAMG145 in statin-intolerant patients. JAMA. 308: 2497–2506.
- [30] K. Fitzgerald, M. Frank-Kamenetsky, S. Shulga-Morskaya, A. Liebow, B.R. Bettencourt, J.R. Sutherland, et al, 2013. Effect of an RNA interference drug on the synthesis of proprotein convertase subtilisin/kexin type 9 (PCSK9) and the concentration of serum LDL cholesterol in healthy volunteers: a randomised, single-blind, placebo-controlled, phase 1 trial. Lancet Jan 4;38: 60–8.
- [31] D.N. Giuseppe, G.Tibolla, A. L. Catapano, 2014. PCSK9 inhibition for the treatment of hypercholesterolemia: Promises and emerging challenges, Vascular Pharmacology 62: 103–111.
- [32] H. Gullberg, M. Rudling, C. Salto, D. Forrest, B. Angelin, B. Vennstrom, 2002. Requirement for thyroid hormone receptor beta in t3 regulation of cholesterol metabolism in mice. Mol Endocrinol, 16: 1767–77.
- [33] K.W. Walton, P.J. Scott, P.W. Dykes, J.W. Davies 1965. The significance of alterations in serum lipids in thyroid dysfunction. II. Alterations of the metabolism and turnover of 131-I-low-density lipoproteins in

hypothyroidism and thyrotoxicosis. Clin Sci, 29:217-38.

- [34] J. Kharlip, D.S. Cooper, 2009. Recent developments in hyperthyroidism. Lancet 373:1930–2.
- [35] R.E. Weiss, Y. Murata, K. Cua, Y. Hayashi, H. Seo, S. Refetoff, 1998. Thyroid hormone action on liver, heart, and energy expenditure in thyroid hormone receptor beta-deficient mice. Endocrinology, 139:4945–52.
- [36] H. Gullberg, M. Rudling, D. Forrest, B. Angelin, B. Vennstrom, 2000. Thyroid hormone receptor betadeficient mice show complete loss of the normal cholesterol 7alpha-hydroxylase (cyp7a) response to thyroid hormone but display enhanced resistance to dietary cholesterol. Mol Endocrinol 14:1739–49.
- [37] L.Z. Huang, H.B. Zhu, 2012. Novel LDL-oriented Pharmacological stretegies. Pharmacol Res 65:402-10.
- [38] G.R. Thompson, 2013. The evidence-base for the efficacy of lipoprotein apheresis in combating cardiovascular disease. Atheroscler Suppl. 14:67–70.
- [39] M.K. Ito, M.P. McGowan, P.M. Moriarty, 2011. Management of familial hypercholesterolemias in adult patients: recommendations from the national lipid association expert panel on familial hypercholesterolemia 5:S38–S45.
- [40] G.F. Watts, D.R. Sullivan, N. Poplawski, et al, 2011. Familial hypercholesterolaemia: a model of care for Australasia. Atherosclerosis Supplements. 12:221–263.
- [41]G.R. Thompson, A. Catapano, S. Saheb, et al, 2010. Severe hypercholesterolaemia: therapeutic goals and eligibility criteria for LDL apheresis in Europe. Curr Opin Lipidol. 21:492–498.
- [42] C. Stefanutti, U. Julius, 2013. Lipoprotein apheresis: state of the art and novelties. Atheroscler Suppl. 14:19–27.
- [43] J.B. Palcoux, M. Atassi-Dumont, P. Lefevre, et al, 2008. Low-density lipoprotein apheresis in children with familial hypercholesterolemia: follow-up to 21 years. Ther Apher Dial. 12:195–201.
- [44] G.R. Thompson, 2008. HEART-UK LDL Apheresis Working Group. Recommendations for the use of LDL apheresis. Atherosclerosis. 198:247–255.
- [45] Thompson GR, Catapano A, Saheb S, et al, 2010. Severe hypercholesterolaemia: therapeutic goals and eligibility criteria for LDL apheresis in Europe. Curr Opin Lipidol. 21:492–498.
- [46] M. Cuchel, E.A. Meagher, H.D.T. Theron, et al, 2013. Efficacy and safety of a microsomal triglyceride transfer protein inhibitor in patients with homozygous familial hypercholesterolaemia: a single-arm, open-label, phase 3 study. Lancet. 381:40–46.
- [47] A. Vogt, K.G. Parhofer, 2013. The potential of mipomersen, an ApoB synthesis inhibitor, to reduce for LDL-apheresis necessity in patients with heterozygous familial hypercholesterolemia and coronary disease. Expert Opin artery Pharmacother.14:691-697.
- [48] G.K. Hovingh, M.H. Davidson, J.J.P Kastelein, A.M. O'Connor, 2013. Diagnosis and treatment of familial hypercholesterolaemia. Eur Heart J. 34:962–971.
- [49] A.S. Wierzbicki, A. Viljoen, T.C. Hardman, D.P. Mikhailidis, 2013. New therapies to reduce low-density lipoprotein cholesterol. Curr Opin Cardiol. 28:452–457.

- [50] M. Moini, P. Mistry, M.L. Schilsky, 2010. Liver transplantation for inherited metabolic disorders of the liver. Curr Opin Organ Tran. 15: 269–276.
- [51] M.H. Nemati, B. Astaneh, 2010. Optimal management of familial hypercholesterolemia: treatment and management strategies. Vasc Health Risk Manag. 6:1079–1088.
- [52] M. Ibrahim, I. El-Hamamsy, M. Barbir, M.H. Yacoub, 2012. Translational lessons from a case of combined heart and liver transplantation for familial hypercholesterolemia 20 years post-operatively. J Cardiovasc Transl Res. 5:351–358.
- [53] H. Buchwald, K.D. Rudser, S.E. Williams, V.N. Michalek, J. Vagasky, J.E. Connett, 2010. Overall mortality, incremental life expectancy, and cause of death at 25 years in the program on the surgical control of the hyperlipidemias. Ann Surg. 251:1034–1040.
- [54] F. Al-Allaf, C. Coutelle, S. Waddington, A. David, R. Harbottle, M. Themis, 2010. LDLR-Gene therapy for familial hypercholesterolaemia: problems, progress, and perspectives. Int Arch Med. 3:36.
- [55] S.H. Kassim, H. Li, P. Bell, et al, 2013. Adenoassociated virus serotype 8 gene therapy leads to significant lowering of plasma cholesterol levels in humanized mouse models of homozygous and heterozygous familial hypercholesterolemia. Hum Gene Ther. 24:19–26.